Author's response to reviews

Title: Psycho-social impact of visual impairment on health-related quality of life among nursing home residents

Authors:

Mahesh Kumar Dev B.Optom (maheshdev2002@gmail.com)
Nabin Paudel PhD student (paudel.nabin@gmail.com)
Niraj Dev Joshi B.Optom (niraidevioshi@gmail.com)
Dev Narayan Shah MS (dn_shah@hotmail.com)
Shishir Subba PhD (shishirsubba21@gmail.com)

Version: 2
Date: 25 February 2014

Author's response to reviews:

To,
Executive Editor,
Christopher Morrey

Dear Sir,

We are submitting a manuscript entitled “Psycho-social impact of visual impairment on health-related quality of life among nursing home residents” for consideration for inclusion in the BMC Health Services Research.

Our manuscript deals with association between visual impairment and health-related quality of life among older adults living in nursing home. The information helps in understanding the impact of VI, psycho-social support and eye care interventions to enhance the health-related quality of life. Thus, promoting the dissemination of knowledge on the health-related quality of life can help in policy making.

I certify on behalf of myself and all co-authors that we have all participated sufficiently in the conception and design of this work and the interpretation of the data (when applicable), as well as in the writing of the manuscript, to take public responsibility for it. Neither this manuscript nor one with substantially similar content under our authorship has been published or is being considered for publication elsewhere.

The institutional review board at the Institute Of Medicine, Tribhuvan University approved the study protocol, and the study followed the tenets of the Declaration of Helsinki.

I certify that any affiliations with or involvement in any organization or entity with a direct financial interest in the subject matter or materials discussed in the manuscript are disclosed in the paper. I declare that we have no conflict of interest.
Yours Sincerely,
Mahesh Kumar Dev, B.Optom
B.P. Koirala Lions Center for Ophthalmic Studies
Department of Ophthalmology
Institute of Medicine, Tribhuvan University
Maharajgunj, Kathmandu
NEPAL.
Cell: +977- 9841633280
Fax No: 977-1-4720142